Author’s response to reviews

Title: Blood Oxygen Level Dependent Magnetic Resonance Imaging for Detecting Pathological Patterns in Lupus Nephritis Patients: A Preliminary Study Using a Decision Tree Model

Authors:

Huilan Shi (shihuilan@vip.126.com)
Zhenfeng Zheng (zhengzhenfeng@vip.126.com)
Junya Jia (youshalei@126.com)
Dong Li (ldno3@163.com)
Li Wei (weilideemail@126.com)
Wenya Shang (wen_xiaoya.5151@163.com)

Version: 2 Date: 13 Dec 2015

Author’s response to reviews:

Respond to editor:

1. Text overlap

Reply: We have revised the overlapped text and eliminated several overlapped sentences.

2. Figure and table

Reply: We uploaded all figures and tables.